Author's response to reviews

Title: Aggressive Local Treatment Containing Intraoperative Radiation Therapy (IORT) for Patients with Isolated Local Recurrences of Pancreatic Cancer: a retrospective analysis

Authors:

Falk FF Roeder (Falk.Roeder@med.uni-heidelberg.de)
Carmen Timke (C.Timke@dkfz.de)
Matthias Uhl (Matthias.Uhl@med.uni-heidelberg.de)
Gregor Habl (gregor.habl@med.uni-heidelberg.de)
Frank W Hensley (Frank.Hensley@med.uni-heidelberg.de)
Markus W Buechler (markus.buechler@med.uni-heidelberg.de)
Robert Krempien (robert.krempien@helios-kliniken.de)
Peter E Huber (p.huber@dkfz.de)
Juergen Debus (juergen.debus@med.uni-heidelberg.de)
Jens Werner (jens.werner@med.uni-heidelberg.de)

Version: 2 Date: 4 April 2012

Author's response to reviews:

Dear Editorial Team,

we have included the requested requirements. In particular we included a statement in the methods section about the approval of our ethics committee and the compliance of the work with the declaration of Helsinki (see page 7) at the end of the methods section.

Best regards

Falk Roeder